

# GeoHealth

### **RESEARCH ARTICLE**

10.1029/2023GH000799

### **Key Points:**

- Temperature-related deaths in the U.S. will increase by a factor of 5 with 3°C of warming, mostly due to aging and increasing population
- Impact of climate change is not significant until 3°C of global warming, and it depends on the level of adaptation beyond that point
- Temperature-related deaths will shift Northward, due to increasing heat-related deaths in poorly heat-adapted Northern cities

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

### Correspondence to:

J. Lee, jangho.lee.92@tamu.edu

### Citation:

Lee, J., & Dessler, A. E. (2023). Future temperature-related deaths in the U.S.: The impact of climate change, demographics, and adaptation. *GeoHealth*, 7, e2023GH000799. https:// doi.org/10.1029/2023GH000799

Received 6 FEB 2023 Accepted 19 JUN 2023

© 2023 The Authors. GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

## Future Temperature-Related Deaths in the U.S.: The Impact of Climate Change, Demographics, and Adaptation

### Jangho Lee<sup>1</sup> <sup>(D)</sup> and Andrew E. Dessler<sup>1</sup> <sup>(D)</sup>

<sup>1</sup>Department of Atmospheric Sciences, Texas A&M University, College Station, TX, USA

**Abstract** Mortality due to extreme temperatures is one of the most worrying impacts of climate change. In this analysis, we use historic mortality and temperature data from 106 cities in the United States to develop a model that predicts deaths attributable to temperature. With this model and projections of future temperature from climate models, we estimate temperature-related deaths in the United States due to climate change, changing demographics, and adaptation. We find that temperature-related deaths increase rapidly as the climate warms, but this is mainly due to an expanding and aging population. For global average warming below 3°C above pre-industrial levels, we find that climate change slightly reduces temperature-related mortality in the U.S. because the reduction of cold-related mortality exceeds the increase in heat-related deaths depends on the level of adaptation. Southern U.S. cities are already well adapted to hot temperatures and the reduction of cold-related mortality. Cities in the Northern U.S. are not well adapted to high temperatures, so the increase in heat-related mortality exceeds the reduction in cold-related mortality. Thus, while the total number of climate-related mortality may not change much, climate change will shift mortality in the U.S. to higher latitudes.

**Plain Language Summary** Deaths due to extreme temperatures is one of the most important impacts of climate change. Here we estimate future temperature related deaths in the US using climate projection, population projection, and adaptation assumptions. We find that in 3°C of global warming level, temperature-related deaths will increase by a factor of five, mostly due to aging and increasing population. Below 3°C warming, impact of climate change slightly decreases temperature-related deaths, because the decrease in cold-related deaths exceeds the increase in heat-related deaths. Above 3°C warming, impact of climate change depends on the level of adaptation. Furthermore, we find that temperature-related mortality shifts Northward. This is due to increase of heat-related deaths in Northern cities, which are currently poorly adapted to heat.

### 1. Introduction

The relationship between temperature and human mortality has been the subject of many previous studies (Berko, 2014; Bobb et al., 2014; Demoury et al., 2022; Dimitrova et al., 2021; Gasparrini & Armstrong, 2011; Gasparrini et al., 2015b; Guo et al., 2011; Kalkstein & Greene, 1997; Ma et al., 2015; Yi & Chan, 2015; Zhang et al., 2016). Previous studies have projected future temperature-related mortality covering different regions, such as global major cities in the U.S. (Anderson et al., 2018; Barreca, 2012; Jackson et al., 2010; Knowlton et al., 2007; Lo et al., 2019; Petkova et al., 2017; Wang et al., 2016; Weinberger et al., 2017), Europe (Hajat et al., 2014; Martínez-Solanas et al., 2021; Muthers et al., 2010), or Asia (Lee & Kim, 2016; Yang et al., 2021). Using historical data sets, previous studies have found that temperature and mortality show a V-shaped curve, where mortality increases as temperatures become very hot or very cold (Berko, 2014; de Schrijver et al., 2022; Dimitrova et al., 2021; Gosling et al., 2009; Vardoulakis et al., 2014).

Prior research has found that climate change will significantly impact future temperature-related mortality; however, these impacts vary depending on location and the implementation of adaptation and mitigation strategies (Gasparrini et al., 2015b, 2017; Lee and Kim, 2016; Martínez-Solanas et al., 2021). For example, Gasparrini et al. (2017) discovered that regions experiencing less intense warming will witness a decline in temperature-related deaths, primarily due to a substantial reduction in cold-related deaths. Conversely, areas subjected to intense warming will see a rise in temperature-related deaths, driven by a considerable increase in heat-related deaths.



(a) Location of 106 US Cities



**Figure 1.** (a) Location of 106 cities used in this study. (b) Risk ratio (RR) of under/over 75 age groups, averaged for all cities in this study. Shaded regions show the 5th percentile to 95th percentile range of RR curve for all cities. RR is the number of deaths at each temperature divided by the number of deaths at the curve's minimum (the minimum mortality temperature), around 22°C.

A particular issue we explore in this paper is the impact of demographics. Older populations are known to be more vulnerable to temperatures extremes (Anderson et al., 2018; Åström et al., 2013; Barnett, 2007; Bobb et al., 2014; de Schrijver et al., 2022; Hintz et al., 2018; Lee & Kim, 2016; Lin et al., 2011; Yi & Chan, 2015; Zhang et al., 2016), and since population is projected to both age and grow globally, the compound effect of demographic and population changes will increase temperature-related mortality (Li et al., 2016; Marsha et al., 2018). Previous studies included demographic and population change in their projection (Deschênes & Greenstone, 2011; Deschenes & Moretti, 2009; Hajat et al., 2014; Jenkins et al., 2014; Lee & Kim, 2016; Li et al., 2016; Petkova et al., 2017; Vardoulakis et al., 2014), mostly using population projections from shared socioeconomic pathways (SSPs) (Hauer, 2019). Utilizing SSPs, prior research has found that the combined effects of a growing and aging population will exacerbate the increase in temperature-related mortality (Anderson et al., 2018).

It is also clear that people will take actions to head off the impacts of extreme temperatures (Barreca et al., 2016; Carson et al., 2006; Davis et al., 2003; Folkerts et al., 2020; Fouillet et al., 2008; Gasparrini et al., 2015a; Kyselý & Plavcová, 2012). However, such adaptation takes resources, which many people do not have. Consequently, how well this can be done is an uncertainty that any analysis of future temperature-related mortality must address.

There are few ways to incorporate adaptation into estimate future temperature related deaths. Previous studies simply shifted the temperature-mortality relationship to warmer temperatures (Folkerts et al., 2020; Gosling et al., 2009; Jenkins et al., 2014), extrapolated the historical trends of temperature-mortality relationship (Muthers et al., 2010; Petkova et al., 2017), or adjusted the slope of temperature-mortality relationship (Jenkins et al., 2014). In this study, we adopt the "analogue city" approach (Heutel et al., 2021; Kalkstein & Greene, 1997; Knowlton et al., 2007), wherein the temperature-mortality model of a city currently experiencing a warmer climate is utilized to establish a mortality model for a presently cooler city under future warming conditions. For example, Knowlton et al. (2007), posited that New York in the 2050s would exhibit a temperature-mortality relationship akin to that of Washington and Atlanta in 1973-1994, as the projected temperatures for New York in the 2050s resemble those of Washington and Atlanta during 1973-1994 period. We have developed a new way of incorporating analog cities into our analysis and we use that to estimate the impact of adaptation.

In this paper, we consider all three of the factors that will impact future temperature-related mortality: climate change, population and demographics change, and adaptation to determine how important each factor is for mortality in the U.S. urban regions.

### 2. Temperature-Mortality Relationship

Mortality data from the National Morbidity Mortality Air Pollution Study (NMMAPS) (Samet et al., 2000) contain the number of daily non-accidental deaths, stratified by age group (<65, 65–75, and >75). We combine the two younger age groups into a single category for ages <75. By this, we have a similar number of deaths in the under 75 and over 75 age groups (49% and 51%, respectively). Data are collected from 106 large U.S. cities (Figure 1a), which contains 65% of the population in the U.S., and cover the period from 1987 to 2000. Population data are also included in NMMAPS, which come from the National Center for Health Statistics. More recent data exhibit limitations, such as being available only on a monthly basis or aggregated at a broader regional level (e.g., state level), which makes them unsuitable for the purpose of this analysis.

Historical hourly 2-m air temperatures from ERA-5 Land reanalysis (Muñoz-Sabater et al., 2021) are averaged to obtain daily average temperatures. The data have a horizontal resolution of  $0.1^{\circ} \times 0.1^{\circ}$  and the average of the nine grid points nearest to the center of each city are used to represent the daily average temperature of the city. Nine grid points encompass an area of approximately  $20 \times 20$  km in the city center, providing a reasonable representation of the city's temperature.

Following the framework of Gasparrini et al. (2015a, 2015b) we use a Distributed Lag Non-Linear Model (DLNM) to describe the association between temperature and mortality. We model the daily number of deaths as a function of daily average temperature separately for each city and age group (under 75 and over 75). An important advantage of the DLNM is that it captures the lagged effect of temperature, where consecutive extreme days results in higher mortality than a single-day event (Gasparrini & Armstrong, 2011; Wang et al., 2016).

Previous studies reported that the impact of a hot day can extend for up to 3 days, while the impact of a cold day could extend 21 days (Demoury et al., 2022; Dimitrova et al., 2021). Therefore, we include lags of up to 21 days in the DLNM model. We also include the day of week to account for the weekly cycle, day of year for the annual cycle, and year for the long-term trend. A detailed explanation of the DLNM model used in this study can be found in Section S1 in Supporting Information S1. Additionally, the sensitivity of including ozone as another confounder in this analysis (Porter and Heald, 2019; Ren et al., 2008) is also discussed in Section S1 in Supporting Information S1.

### 3. Historical Temperature-Related Mortality

Figure 1b summarizes the mortality risk as a function of temperature, averaged over all cities (curves for 25 most populated cities can be found in Figure S1 in Supporting Information S1). The quantity plotted here, the cumulative risk ratio (RR), is the number of deaths at each temperature divided by the number of deaths at the minimum mortality temperature (MMT), after summing the RR at each lag, up to 21 days. Our curve is similar to those found in previous work (Gasparrini et al., 2015a, 2015b, 2017; Guo et al., 2011; Lin et al., 2011; Ma et al., 2015; Yi & Chan, 2015; Zhang et al., 2016).

With the regression models for each city and age group, we calculate the number of temperature-related excess deaths in the NMMAPS data in two steps. First, we define baseline deaths, which is the number of deaths at the MMT, calculated by averaging the number of deaths at temperatures around MMT ( $\pm 0.5^{\circ}$ C). With this baseline death value, we then calculate the number of deaths in each city using observed temperatures and the mortality-temperature curves.

There are an average of 36,444 temperature-related deaths per year between 1987 and 2000 (solid line in Figure 2a). There is a clear trend over this period, and we can remove the effect of population changes by dividing the number of excess deaths by the population in each year, and then multiplying by the average population over the period. Doing this removes most of the trend (dashed line in Figure 2a).

We will separate temperature-related deaths occurring above and below MMT, which by convention we refer to as heat- and cold-related deaths. We estimate there are an average of 4,819 heat-related deaths per year and 31,625 cold-related deaths. This is consistent with previous work that also found that most of the deaths were due to cold, rather than heat (Vardoulakis et al., 2014). Upon examining the age distribution of temperature-related deaths in each city, we found that, on average, 75.3% of deaths are attributed to the older age group (over 75), with an inter-city standard deviation equal to 6.2%. The older age group accounts for 75.6% ( $1\sigma = 4.6\%$ ) of heat-related mortality and 75.1% ( $1\sigma = 6.9\%$ ) of cold-related mortality, despite comprising only 5.1% ( $1\sigma = 1.3\%$ ) of the population. This fact that temperature-related mortality is mainly in the older age groups will be important later in the paper.

Although 86% of temperature-related deaths are due to cold-related mortality (4,819 heat-related deaths, 31,625 cold-related deaths), most of these "cold-related" deaths occur at temperatures only slightly below the MMT, which is typically around 22°C. For example, 26% of cold-related mortality is occurring within 5°C of the MMT, with an inter-city standard deviation of 11%. While the risk of temperature-related death at these pleasant temperatures is low, their frequent occurrence still contributes to a significant number of deaths, a point also highlighted by Gasparrini et al. (2015a, 2015b).



### GeoHealth



Figure 2. Time series of temperature-related deaths across all 106 cities. (a) The solid line represents total temperature-related deaths, while the dashed line indicates temperature-related deaths with a fixed population (average population from the 1987–2000 period). (b) Similar to (a), but for heat-related deaths. (c) Similar to (a), but for cold-related deaths. (d) Time series of deaths occurring in temperatures above the 97.5th percentile. (e) Similar to (d), but for temperatures below the 2.5th percentile.

This led us to investigate mortality resulting from extreme heat and cold. For each city, we selected days with temperatures exceeding the 97.5th percentile (calculated from the 1987–2000 temperatures) and falling below the 2.5th percentile (Gasparrini et al., 2015b), which we refer to as extreme heat- and cold-related deaths (Figures 2d and 2e). When considering all cities, there are, on average, 1,022 deaths per year attributable to extreme heat and 1,922 to extreme cold, representing 21% and 6% of total heat and cold-related deaths, respectively. This indicates that heat-related deaths tend to be more concentrated around extreme heat, while cold-related deaths exhibit a less pronounced skew toward extreme cold.

### 4. Adaptation

We know that people will adapt as much as possible to extreme heat. However, adaptation takes resources, and it is unclear the extent to which successful adaptation is achievable. Therefore, we analyze two limiting adaptation scenarios. In the first scenario, we assume no further adaptation, with the relative risk (RR) curve for each city remaining constant based on the 1987–2000 mortality data as the climate warms (Figure S1 in Supporting Information S1). This is the no-adaptation scenario.

Our second scenario represents a strong adaptation case, in which cities adapt effectively to the warmer temperatures. We begin this calculation by quantifying how well adapted each city is in the 1987–2000 climate by calculating the linear slope of each city's cumulative RR versus temperature for temperatures above the MMT (hot RR slope) and slope of RR below the MMT (cold RR slope). While the RR curves are not linear, the linear fit is a metric for how steeply the curve rises. We do this fit separately for each age group. Figures 3a and 3c show the hot RR slope of all of the cities regressed against median of daily average temperatures of that city's hot season (June, July, and August, JJA) over the 1987–2000 period. Figures 3b and 3d show the cold RR slope regressed against the median daily average temperature of the cold season (December, January, and February, DJF).

There is a clear anti-correlation between the RR slopes and seasonal temperatures. Cities with warmer summers are less vulnerable to heat-related mortality (hot RR slopes are lower, Figures 3a and 3c), while cities with colder winters are less vulnerable to cold-related mortality (cold RR slopes are lower, Figures 3b and 3d). One





**Figure 3.** Relationship between the slope of each city's RR curve and that city's climate. (a) Relationship between slope of RR curve above minimum mortality temperature (MMT) (hot RR slope) for under 75 age groups and the JJA median daily temperature. The points represent individual cities, and the line is a linear regression fit. (b) Same as (a), but for slope of RR curve below MMT (cold RR slope) and the median DJF temperature. (c, d) Same as (a, b), but for over 75 age groups.

can also think of these fit lines in Figure 3 as measures of how humans adapt to changes in climate (Gasparrini et al., 2015b; Heutel et al., 2021; Kalkstein & Greene, 1997; Knowlton et al., 2007).

The regression lines in Figure 3 between temperature and slope of cities' RR curves allows us to parameterize how the RR curves change as temperatures warm. In our strong adaptation assumption, we assume that each cities' RR slope evolves following the regression lines in Figure 3, using each city's temperature over the previous 10 years to estimate that year's RR slope. Hot cities such as Miami, Houston, and Phoenix will warm past the data points in Figure 3, that is, they lack suitable analogs in a warmer climate. For those cities, we simply extrapolate the regression lines into the warmer climates, although we do not allow the slope to decline below zero. Cities that are already hot and well-adapted to high temperatures are affected by this constraint. At 3°C global warming, the hot side RR slope (Figures 3a and 3c) falls below zero in Austin (TX), El Paso (TX), Houston (TX), Los Angeles (CA), Miami (FL), Phoenix (AZ), San Antonio (TX), and Tucson (AZ) (eight cities). The impact of constraining hot side RR slope not to fall below zero will be discussed later in the paper. We account for adaptation on the cold side by scaling the cold RR slope by 18.5% of the ratio used to scale the hot RR slope (18.5% represents the ratio of the hot-side to cold-side fits shown in Figure 3). Having established the RR slopes for both no adaptation (points in Figure 3) and strong adaptation (modified RR slope due to temperature metric change), we can estimate the impact of adaptation on temperature-related mortality. A detailed example of incorporating adaptation is provided in Section S3 in Supporting Information S1.

Another way we could have included adaptation is to shift the MMT toward warmer temperatures as the climate warms. This will reduce deaths due to warm temperatures, but how it affects cold mortality depends on what assumption is made for how the cold side of the RR curve evolves as the MMT shifts. If one just translates the RR curve as the MMT shifts, then cold mortality increases, largely offsetting the benefits of reduced warm mortality. Other assumptions are of course possible. Since we don't know what the best assumption is, we decided to not

Description of NA-CORDEX Members Used in This Study			
Scenario	Global climate model	Regional climate model	Bias-correction
Historical + RCP 8.5	CanESM2	CanRCM4	MBCn using Daymet
		CRCM5-UQAM	
	GEMatm-Can	CRCM5-UQAM	
	GEMatm-MPI	CRCM5-UQAM	
	GFDL-ESM2M	RegCM4	
		WRF	
	HadGEM2-ES	RegCM4	
		WRF	
	MPI-ESM-LR	CRCM5-UQAM	
		RegCM4	
		WRF	
	MPI-ESM-MR	CRCM5-UOAM	

Table 1

adjust the MMT as the climate warms. In the end, our preliminary calculations suggest that this decision does not impact the conclusions of the paper.

### 5. Future Temperature-Related Deaths

For our projection of future temperature-related mortality, we utilize historical and RCP 8.5 scenario runs from NA-CORDEX (Mearns et al., 2017), which contain bias-corrected outputs of regional climate model runs over North America, using boundary conditions from global climate models (GCM). Twelve combinations of GCMs and RCMs are used in this study, and these are summarized in Table 1. Historical simulations cover the period from 1950 to 2005 and RCP 8.5 simulations cover 2006 to 2099. The NA-CORDEX data have a  $0.22^{\circ} \times 0.22^{\circ}$ horizontal resolution, so we use the four grid points nearest to each city to represent the city's temperature. This covers approximately a  $22 \times 22$  km area in the city center, which is consistent with the ERA-5 temperatures used to develop the temperature-mortality model.

We validate the NA-CORDEX ensemble by predicting temperature-related deaths in 1987–2000 period. We do this by plugging temperatures from the NA-CORDEX ensemble for each city over that period into that city's regression model. The average number of temperature-related deaths estimated using NA-CORDEX temperatures is 36,675 (inter-model 95% CI = 36,189-37,231), in which 5,067 (95% CI = 4,666-5,332) deaths are heat-related, and 31,608 (95% CI = 31,111–32,331) are cold-related. Using ERA-5 temperatures, we estimated 36,444, 4,819, and 31,625 deaths, respectively. This provides some confidence in the NA-CORDEX temperature fields.

For future temperature-related mortality predictions, we also need predictions of population and demographics. For this, we use the SSP5 scenario, a fossil-fueled development scenario, which corresponds with the RCP8.5 emissions scenario (Juzbašić et al., 2022; Rohat et al., 2021; Roy et al., 2021), which is driving the NA-CORDEX models. We use data from Hauer (2019), which contains county-level estimates of population and demographics at 5-year intervals from 2020 to 2100. To convert the county-level estimate to the city level, we extract counties containing each city in our analysis. 74% of the cities are within one county, and for these we assume that the city's population remains a constant fraction of the county's population. For cities that are in multiple counties, we sum the population of all counties that include the cities. Because the counties and city do not perfectly overlap, we take historical demographic data from 2020 and SSP5 data from 2020 and estimate the fraction of the total counties' population living in the city, and assume that fraction is constant over the century. From this, we come up with time series of population estimates in two age groups for each city in the coming century. A summary of population and demographic projections, as well as sensitivity due to choice of socioeconomic pathways can be found on the Section S4 in Supporting Information S1.

To estimate future deaths, we plug NA-CORDEX temperatures for the 21st century for each city into that city's regression model (Figure S1 in Supporting Information S1) and then use population and demographic information to convert the RR values to temperature-related mortality numbers.

RCP 8.5 emissions will likely exceed actual emissions, so we plot estimated mortality as a function of global average surface temperature (relative to the 1850–1859 period, before significant human-induced warming occurred). We estimate global average warming in each year of the NA-CORDEX from averages of the four GCM included in NA-CORDEX: CanESM2 (5 ensemble members) (Chylek et al., 2011), GFDL-ESM2M (1 run) (Dunne et al., 2020), HadGEM2-ES (3 ensemble members) (Collins et al., 2011), and MPI-ESM (100 ensemble members) (Maher et al., 2019) with historical and RCP 8.5 forcing. To do this, we first average the ensemble members of each climate model and then average those to come up with a single global average temperature time series. This gives us the global average warming of 0.83 in 2000 and 1.37°C in year 2022, close to observed values in ERA-5. We did not incorporate other emission scenarios (e.g., RCP 4.5 and RCP 2.6) in this study because the NA-CORDEX database lacks sufficient model runs for those scenarios.

Most cities will warm more than the global average. A global temperature increase of 3°C relative to pre-industrial times corresponds to 2°C warming above the 2012–2022 period. The 106 major cities analyzed here are expected to warm by 2.41°C relative to 2012–2022, 20% more than the global average. Cities further from the coast and at higher latitude are expected to warm up more. The top three cities with the highest projected temperature increases are Minneapolis, MN (0.96°C larger than the global average), Milwaukee, WI (0.88°C), and Muskegon, MI (0.86°C). Detailed city-by-city data are available in Table S2 in Supporting Information S1.

With future climate projections from NA-CORDEX, future population and demographics projection from SSP5, and our two adaptation scenarios, we calculate future temperature-related mortality (Figures 4a–4d). Looking at total temperature-related deaths with no adaptation, we find that there are 45,800 deaths annually between 2011 and 2020 (1.16°C warming) and that is projected to grow to 200,000 annually with 3°C of global average warming, with both heat- and cold-related deaths increasing (Figures 4a and 4b). Adaptation will decrease this number, reducing the increase of temperature-related mortality at 3°C by about 28%. There are 12,500 deaths annually between 2011 and 2020 due to extreme temperatures (Figures 4c and 4d), which is projected to increase to 63,000 annually at 3°C, a proportionally larger increase than all-temperature deaths.

We now decompose the increase in temperature-related mortality into contributions from climate change, demographics change, and population change. To estimate the impact of each of these terms, we repeat the mortality calculation with that term fixed and then subtract the values obtained from the calculation with all terms varying.

To estimate the impact of climate change, we fix climate by repeating the daily temperature of recent years (2011–2020) for the entire period (1987–2100) and then subtracting the resulting temperature-related mortality from the all-factor calculation. For the no-adaptation case, lives saved by less cold balances the lives lost due to more hot temperatures until about 3°C. Above that, the increase in heat-related mortality overwhelms and total mortality rises rapidly. For the strong adaptation scenario, temperature-related mortality decreases at all temperatures. Allowing the RR slope to decrease below zero results in a 50% reduction in the number of temperature-related deaths in eight cities that were restricted by the constraint, corresponding to an annual decrease of 2,192 deaths at a 3°C warming level. This represents a 1.1% decrease in total temperature-related deaths nationally, indicating that it has a minimal effect on our overall findings.

We also find that temperature-related mortality in response to extreme temperatures will increase at all levels of warming (Figures 4g and 4h). This tells us that most of the lives saved in a warming world are due to a reduction in moderate cold temperatures.

Next, we look at impact of demographics (Figures 4i-41) by performing a fixed-demographics calculation that fixes the ratio of under/over 75 population to the 2011–2020 average and then subtracting this from the all-factor calculation. We find that the aging of our population drives an enormous increase in deaths (Figure 4i) due to the older age group being more vulnerable to temperature-related mortality (Figure 4j).

Finally, we calculate the impact of population (Figures 4m-4p) by fixing population at the 2011–2020 average value and subtracting the results from the all-factor calculation. As the population increases, the total number of deaths also increases.





**Figure 4.** Estimates of future temperature-related deaths as a function of global average warming. (a–d) Future temperature-related deaths incorporating all factors: climate, demographics, and population. Upper limit of shaded region represents no-adaptation scenario, while the lower limit represents the adaptation scenario. (a) All temperature related mortality, (b) heat- and cold-related deaths, (c) mortality due to extreme temperatures, (d) mortality due to extreme heat and cold. Lower rows follow the same pattern as (a–d), but considering only climate change (e–h), demographics change (i–l), and population change (m–p). In all panels, dashed lines represent the average of the current value (2011–2020).

Comparing the three contributing terms, we find that changes in demographics and population are the most important driver of future mortality, and then climate change. This likely reflects the enormous investments in adaptation that have already been made (e.g., nearly 100% air conditioner penetration in cities like Phoenix and Houston). These results stand in contrast to analyses of poorer countries, which show large increases in temperature-related mortality as the climate warms in the future (Carleton et al., 2022).

### 6. The Spatial Pattern of Temperature-Related Deaths

We now analyze the spatial distribution of heat-related mortality. We focus on the meridional variations in number of deaths at 3°C global average warming, approximately business-as-usual warming for 2100. We find that most of the temperature-related deaths occur between 40° and 45°N (Figure 5a). Analyzing per capita deaths, we find they are also weighted toward higher latitudes (Figure 5c).

Looking at the climate contribution (Figures 5e and 5g), we see that climate change shifts mortality poleward: decreasing deaths at lower latitudes and potentially increasing them (depending on adaptation) at higher latitudes. This occurs because Southern cities in the U.S. are already well adapted to heat (Figure 3), so further warming does not add significantly to heat-related deaths. However, these Southern cities do experience a decline of cold-related deaths, leading to a net reduction in temperature-related mortality. Northern cities, on the other hand,





**Figure 5.** Meridional distribution of temperature-related deaths in 3°C world. (a) Number of temperature related deaths in 3°C world. The upper limit of the shaded region represents no-adaptation scenario, while the lower limit represents the adaptation scenario. (b) Same as (a), but for heat- and cold-related deaths. (c, d) Same as (a, b), but per capita (each bin has been divided by population in that bin). (e–h) Contribution of climate change to mortality, (i–l) contribution of demographic changes to mortality, (m–n) contribution of changes in population.

are less adapted to heat, so they experience large increases in heat-related mortality, which exceeds the decline in cold-related mortality (Carleton et al., 2022; Knowlton et al., 2007).

The impact of adaptation is particularly pronounced between  $30^{\circ}$  and  $35^{\circ}N$  (Figure 5e) due to demographics. Currently, the  $30^{\circ}-35^{\circ}N$  region is the second youngest region (percentage of over 75 age group = 5.29%); when the Earth reaches  $3^{\circ}C$  of global average warming, it will be the oldest region (17.32%). Since the older age group is both more vulnerable to high temperatures and more sensitive to adaptation (Figure 3), adaptation will have a large impact on mortality over this latitude range.

We have made similar plots for extreme heat- and cold-related deaths (mortality in the hottest and coldest 30 days), and they show a stronger impact from climate change (Figure S4 in Supporting Information S1). Numbers for all temperature-related deaths at 3°C warming are tabulated in Section S6 in Supporting Information S1.

### 7. Conclusions

In this paper, we use mortality and temperature data obtained between 1987 and 2000 to develop a temperature-mortality relationship for 106 cities in the U.S. covering about 65% of the total population. We then combine these regression models with temperatures from an ensemble of high-resolution climate simulations to estimate future temperature-related deaths. Because of the key role of adaptation, we make two different adaptation scenarios: a scenario with no adaptation and what we consider to be an aggressive adaptation scenario

that follows the observed variations in adaptation between cities with different climates. We also incorporate estimates of changing population and its age distribution.

We estimate that there was an average of 36,444 temperature-related deaths per year during the period 1987–2000 in the cities in our data set. Consistent with previous work (Berko, 2014; Gasparrini et al., 2015a, 2017; Heutel et al., 2021), we find that 86% of these deaths were cold-related. Most of the cold-related deaths took place at moderate temperatures just below the MMT, typically around 20–22°C, so they are categorized as cold related even though many would consider the temperatures to be mild.

We project that, with a warming climate and an increasing and aging population, temperature-related deaths in these cities will reach 200,000 per year at 3°C of global average warming without adaptation. Adaptation reduces the increase of this number of temperature-related deaths at 3°C of warming by 28%.

By decomposing mortality into climate, demographics, and population factors, we find that demographic shifts, primarily the aging of the population, and increasing population—will be the biggest drivers of increased temperature-related mortality. Climate change will cause small changes in mortality below 3°C of global average warming due to offsetting decreases in cold-related mortality and increases in heat-related deaths. Above 3°C, the result depends on the level of adaptation with increases in heat-related deaths dominating without adaptation. Without adaptation, total mortality rises rapidly; with adaptation, mortality declines. In our strong adaptation scenario, total heat-related mortality decreases at all levels of warming.

The health impacts of extreme heat are just one of many health impacts of climate change (Rocque et al., 2021). As such, this research should not be interpreted as suggesting that climate change will yield net health benefits for low levels of warming.

While changes in temperature-related mortality due to climate change may be small below 3°C, there is a meridional shift of mortality, with deaths shifting from the South to the North (Figures 5g and 5h). Since Southern cities in U.S. are already well adapted to heat, additional warming does not add a significant number of deaths. Northern cities, on the other hand, are not well adapted to heat, so heat-related mortality increases there dominate decreases in cold-related mortality.

Ultimately, no one knows how effectively we will adapt to the warmer temperatures of the coming century. However, the investments society has made to make cities like Houston or Phoenix livable in a hot climate are massive and it is far from assured that we will make similar investments in other cities as the climate warms. Many adaptive responses (e.g., installing air conditioning, improved health care, better urban planning) are too expensive for poorer individuals or communities, so adaptation will necessarily require society to pay for much of the adaptation. This would represent a huge transfer of wealth from richer to poorer members of our society, a dicey proposition in today's political environment.

There are important limitations to our analysis. Among the most important, our analysis covered 106 medium and large cities in the U.S., so we can't reach any conclusions about rural populations of the U.S. population or states that are not included in the mortality data set (MT, ID, WY, ND, and SD). Second, we cannot comment on the future of heat-related mortality in the rest of the world. However, given the wealth of the U.S., our present levels of adaptation are higher than in many poorer countries and our ability to enhance our adaptation is also higher. Thus, it seems likely that heat-related mortality will be a more significant problem in the rest of the world as climate change progresses through the century (Carleton et al., 2022). Finally, a limitation of our analysis arises from the temperature data set and projections used. We utilized  $22 \times 22$  km boxes from the ERA-5 temperature data set to represent the city's temperatures and sensitivity tests indicate that the size of the box we use introduces little error in our analysis. However, there are clear limitations to using this type of reanalysis for estimating the temperature of the city, in particular the neglect of the urban heat island effect. Future analysis should focus on using higher resolution temperature data in analyses to better capture this effect.

Nevertheless, this study integrates climate change, population and demographic change, and adaptation to project future temperature-related deaths in the U.S. To our knowledge, only a handful of studies have attempted such a comprehensive approach, particularly emphasizing the range of adaptation. By doing so, our study highlights the importance of adaptation measures and identifies the most vulnerable cities in terms of future temperature-related deaths, in addition to considering climate and population/demographic changes.



### **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

### **Data Availability Statement**

Mortality data from National Morbidity Mortality Air Pollution Study is available on R package: https:// cran.r-project.org/src/contrib/Archive/NMMAPSlite/. Historical ERA-5 temperature data are available on Climate Data Store: https://cds.climate.copernicus.eu/cdsapp#!/dataset/reanalysis-era5-land?tab=overview. NA-CORDEX temperature data are available on Climate Data Gateway by NCAR: https://www. earthsystemgrid.org/search/cordexsearch.html. Future SSP population and demographics data are available in Socioeconomic Data and Applications Center (SEDAC): https://sedac.ciesin.columbia.edu/data/set/ ssp-1-8th-urban-land-extent-projection-base-year-ssp-2000-2100.

### References

- Anderson, G. B., Oleson, K. W., Jones, B., & Peng, R. D. (2018). Projected trends in high-mortality heatwaves under different scenarios of climate, population, and adaptation in 82 US communities. *Climatic Change*, 146(3–4), 455–470. https://doi.org/10.1007/s10584-016-1779-x Åström, D. O., Forsberg, B., Edvinsson, S., & Rocklöv, J. (2013). Acute fatal effects of short-lasting extreme temperatures in Stockholm, Sweden: Evidence across a century of change. *Epidemiology*, 24(6), 820–829. https://doi.org/10.1097/01.ede.0000434530.62353.0b
- Barnett, A. G. (2007). Temperature and cardiovascular deaths in the US elderly: Changes over time. *Epidemiology*, *18*(3), 369–372. https://doi. org/10.1097/01.ede.0000257515.34445.a0
- Barreca, A. I., Clay, K., Deschenes, O., Greenstone, M., & Shapiro, J. S. (2016). Adapting to climate change: The remarkable decline in the US temperature-mortality relationship over the twentieth century. *Journal of Political Economy*, 124(1), 105–159. https://doi.org/10.1086/684582
- Barreca, A. I. (2012). Climate change, humidity, and mortality in the United States. *Journal of Environmental Economics and Management*, 63(1), 19–34. https://doi.org/10.1016/j.jeem.2011.07.004
- Berko, J. (2014). Deaths attributed to heat, cold, and other weather events in the United States, 2006–2010. US Department of Health and Human Services, Centers for Disease Control and Prevention.
- Bobb, J. F., Peng, R. D., Bell, M. L., & Dominici, F. (2014). Heat-related mortality and adaptation to heat in the United States. *Environmental Health Perspectives*, 122(8), 811–816. https://doi.org/10.1289/ehp.1307392
- Carleton, T., Jina, A., Delgado, M., Greenstone, M., Houser, T., Hsiang, S., et al. (2022). Valuing the global mortality consequences of climate change accounting for adaptation costs and benefits. *Quarterly Journal of Economics*, 137(4), 2037–2105. https://doi.org/10.1093/qje/qjac020
- Carson, C., Hajat, S., Armstrong, B., & Wilkinson, P. (2006). Declining vulnerability to temperature-related mortality in London over the 20th century. American Journal of Epidemiology, 164(1), 77–84. https://doi.org/10.1093/aje/kwj147
- Chylek, P., Li, J., Dubey, M., Wang, M., & Lesins, G. (2011). Observed and model simulated 20th century Arctic temperature variability: Canadian Earth system model CanESM2. *Atmospheric Chemistry and Physics Discussions*, *11*, 22893–22907. https://doi.org/10.5194/ acpd-11-22893-2011
- Collins, W., Bellouin, N., Doutriaux-Boucher, M., Gedney, N., Halloran, P., Hinton, T., et al. (2011). Development and evaluation of an Earth-System model–HadGEM2. *Geoscientific Model Development*, *4*, 1051–1075. https://doi.org/10.5194/gmd-4-1051-2011
- Davis, R. E., Knappenberger, P. C., Michaels, P. J., & Novicoff, W. M. (2003). Changing heat-related mortality in the United States. Environmental Health Perspectives, 111(14), 1712–1718. https://doi.org/10.1289/ehp.6336
- Demoury, C., Aerts, R., Vandeninden, B., Van Schaeybroeck, B., & De Clercq, E. M. (2022). Impact of short-term exposure to extreme temperatures on mortality: A multi-city study in Belgium. *International Journal of Environmental Research and Public Health*, 19(7), 3763. https:// doi.org/10.3390/ijerph19073763
- Deschênes, O., & Greenstone, M. (2011). Climate change, mortality, and adaptation: Evidence from annual fluctuations in weather in the US. American Economic Journal: Applied Economics, 3(4), 152–185. https://doi.org/10.1257/app.3.4.152
- Deschenes, O., & Moretti, E. (2009). Extreme weather events, mortality, and migration. *The Review of Economics and Statistics*, 91(4), 659–681. https://doi.org/10.1162/rest.91.4.659
- de Schrijver, E., Bundo, M., Ragettli, M. S., Sera, F., Gasparrini, A., Franco, O. H., & Vicedo-Cabrera, A. M. (2022). Nationwide analysis of the heat-and cold-related mortality trends in Switzerland between 1969 and 2017: The role of population aging. *Environmental Health Perspectives*, *130*, 037001. https://doi.org/10.1289/EHP9835
- Dimitrova, A., Ingole, V., Basagana, X., Ranzani, O., Mila, C., Ballester, J., & Tonne, C. (2021). Association between ambient temperature and heat waves with mortality in South Asia: Systematic review and meta-analysis. *Environment International*, 146, 106170. https://doi. org/10.1016/j.envint.2020.106170
- Dunne, J., Horowitz, L. W., Adcroft, A. J., Ginoux, P., Held, I. M., John, J. G., et al. (2020). The GFDL Earth System Model version 4.1 (GFDL-ESM 4.1): Overall coupled model description and simulation characteristics. *Journal of Advances in Modeling Earth Systems*, 12(11), e2019MS002015. https://doi.org/10.1029/2019ms002015
- Folkerts, M. A., Bröde, P., Botzen, W., Martinius, M. L., Gerrett, N., Harmsen, C. N., & Daanen, H. A. (2020). Long term adaptation to heat stress: Shifts in the minimum mortality temperature in The Netherlands. *Frontiers in Physiology*, *11*, 225. https://doi.org/10.3389/fphys.2020.00225
- Fouillet, A., Rey, G., Wagner, V., Laaidi, K., Empereur-Bissonnet, P., Le Tertre, A., et al. (2008). Has the impact of heat waves on mortality changed in France since the European heat wave of summer 2003? A study of the 2006 heat wave. *International Journal of Epidemiology*, 37(2), 309–317. https://doi.org/10.1093/ije/dym253
- Gasparrini, A., & Armstrong, B. (2011). The impact of heat waves on mortality. *Epidemiology*, 22(1), 68–73. https://doi.org/10.1097/ede.0b013e3181fdcd99
- Gasparrini, A., Guo, Y., Hashizume, M., Kinney, P. L., Petkova, E. P., Lavigne, E., et al. (2015a). Temporal variation in heat–mortality associations: A multicountry study. *Environmental Health Perspectives*, 123(11), 1200–1207. https://doi.org/10.1289/ehp.1409070

### Acknowledgments

This work was supported by NSF Grant AGS-1841308 to Texas A&M University.

- Gasparrini, A., Guo, Y., Hashizume, M., Lavigne, E., Zanobetti, A., Schwartz, J., et al. (2015b). Mortality risk attributable to high and low ambient temperature: A multicountry observational study. *The Lancet*, 386(9991), 369–375. https://doi.org/10.1016/s0140-6736(14)62114-0
  Gasparrini, A., Guo, Y., Sera, F., Vicedo-Cabrera, A. M., Huber, V., Tong, S., et al. (2017). Projections of temperature-related excess mortality under climate change scenarios. *The Lancet Planetary Health*, 1(9), e360–e367. https://doi.org/10.1016/s2542-5196(17)30156-0
- Gosling, S. N., McGregor, G. R., & Lowe, J. A. (2009). Climate change and heat-related mortality in six cities Part 2: Climate model evaluation and projected impacts from changes in the mean and variability of temperature with climate change. *International Journal of Biometeorology*, 53(1), 31–51. https://doi.org/10.1007/s00484-008-0189-9
- Guo, Y., Barnett, A. G., Pan, X., Yu, W., & Tong, S. (2011). The impact of temperature on mortality in Tianjin, China: A case-crossover design with a distributed lag nonlinear model. *Environmental Health Perspectives*, 119(12), 1719–1725. https://doi.org/10.1289/ehp.1103598
- Hajat, S., Vardoulakis, S., Heaviside, C., & Eggen, B. (2014). Climate change effects on human health: Projections of temperature-related mortality for the UK during the 2020s, 2050s and 2080s. *Journal of Epidemiology & Community Health*, 68(7), 641–648. https://doi.org/10.1136/ jech-2013-202449
- Hauer, M. E. (2019). Population projections for US counties by age, sex, and race controlled to shared socioeconomic pathway [Dataset]. Scientific Data, 6, 1–15. https://doi.org/10.1038/sdata.2019.5
- Heutel, G., Miller, N. H., & Molitor, D. (2021). Adaptation and the mortality effects of temperature across US climate regions. *The Review of Economics and Statistics*, 103, 740–753. https://doi.org/10.1162/rest\_a\_00936
- Hintz, M. J., Luederitz, C., Lang, D. J., & von Wehrden, H. (2018). Facing the heat: A systematic literature review exploring the transferability of solutions to cope with urban heat waves. Urban Climate, 24, 714–727. https://doi.org/10.1016/j.uclim.2017.08.011
- Jackson, J. E., Yost, M. G., Karr, C., Fitzpatrick, C., Lamb, B. K., Chung, S. H., et al. (2010). Public health impacts of climate change in Washington state: Projected mortality risks due to heat events and air pollution. *Climatic Change*, 102(1–2), 159–186. https://doi.org/10.1007/ s10584-010-9852-3
- Jenkins, K., Hall, J., Glenis, V., Kilsby, C., McCarthy, M., Goodess, C., et al. (2014). Probabilistic spatial risk assessment of heat impacts and adaptations for London. *Climatic Change*, 124(1-2), 105–117. https://doi.org/10.1007/s10584-014-1105-4
- Juzbašić, A., Ahn, J. B., Cha, D. H., Chang, E. C., & Min, S. K. (2022). Changes in heat stress considering temperature, humidity, and wind over East Asia under RCP8.5 and SSP5-8.5 scenarios. *International Journal of Climatology*, 42(12), 6579–6595. https://doi.org/10.1002/joc.7636
- Kalkstein, L. S., & Greene, J. S. (1997). An evaluation of climate/mortality relationships in large US cities and the possible impacts of a climate change. *Environmental Health Perspectives*, 105(1), 84–93. https://doi.org/10.1289/ehp.9710584
- Knowlton, K., Lynn, B., Goldberg, R. A., Rosenzweig, C., Hogrefe, C., Rosenthal, J. K., & Kinney, P. L. (2007). Projecting heat-related mortality impacts under a changing climate in the New York City region. *American Journal of Public Health*, 97(11), 2028–2034. https://doi. org/10.2105/ajph.2006.102947
- Kyselý, J., & Plavcová, E. (2012). Declining impacts of hot spells on mortality in the Czech republic, 1986–2009: Adaptation to climate change? *Climatic Change*, 113(2), 437–453. https://doi.org/10.1007/s10584-011-0358-4
- Lee, J. Y., & Kim, H. (2016). Projection of future temperature-related mortality due to climate and demographic changes. *Environment Interna*tional, 94, 489–494. https://doi.org/10.1016/j.envint.2016.06.007
- Li, T., Horton, R. M., Bader, D. A., Zhou, M., Liang, X., Ban, J., et al. (2016). Aging will amplify the heat-related mortality risk under a changing climate: Projection for the elderly in Beijing, China. Scientific Reports, 6, 1–9. https://doi.org/10.1038/srep28161
- Lin, Y.-K., Ho, T.-J., & Wang, Y.-C. (2011). Mortality risk associated with temperature and prolonged temperature extremes in elderly populations in Taiwan. Environmental Research, 111(8), 1156–1163. https://doi.org/10.1016/j.envres.2011.06.008
- Lo, Y. E., Mitchell, D. M., Gasparrini, A., Vicedo-Cabrera, A. M., Ebi, K. L., Frumhoff, P. C., et al. (2019). Increasing mitigation ambition to meet the Paris Agreement's temperature goal avoids substantial heat-related mortality in US cities. *Science Advances*, 5(6), eaau4373. https:// doi.org/10.1126/sciadv.aau4373
- Ma, W., Wang, L., Lin, H., Liu, T., Zhang, Y., Rutherford, S., et al. (2015). The temperature–mortality relationship in China: An analysis from 66 Chinese communities. *Environmental Research*, 137, 72–77. https://doi.org/10.1016/j.envres.2014.11.016
- Maher, N., Milinski, S., Suarez-Gutierrez, L., Botzet, M., Dobrynin, M., Kornblueh, L., et al. (2019). The Max Planck Institute Grand Ensemble: Enabling the exploration of climate system variability. *Journal of Advances in Modeling Earth Systems*, 11(7), 2050–2069. https://doi.org/10.1029/2019ms001639
- Marsha, A., Sain, S., Heaton, M., Monaghan, A., & Wilhelmi, O. (2018). Influences of climatic and population changes on heat-related mortality in Houston, Texas, USA. *Climatic Change*, 146(3–4), 471–485. https://doi.org/10.1007/s10584-016-1775-1
- Martínez-Solanas, È., Quijal-Zamorano, M., Achebak, H., Petrova, D., Robine, J. M., Herrmann, F. R., et al. (2021). Projections of temperature-attributable mortality in Europe: A time series analysis of 147 contiguous regions in 16 countries. *The Lancet Planetary Health*, 5(7), e446–e454. https://doi.org/10.1016/s2542-5196(21)00150-9
- Mearns, L., McGinnis, S., Korytina, D., Arritt, R., Biner, S., Bukovsky, M., et al. (2017). The NA-CORDEX dataset, version 1.0 [Dataset]. NCAR Climate Data Gateway. https://doi.org/10.5065/D6SJ1JCH
- Muñoz-Sabater, J., Dutra, E., Agustí-Panareda, A., Albergel, C., Arduini, G., Balsamo, G., et al. (2021). ERA5-Land: A state-of-the-art global reanalysis dataset for land applications [Dataset]. Earth System Science Data, 13(9), 4349–4383. https://doi.org/10.5194/essd-13-4349-2021
- Muthers, S., Matzarakis, A., & Koch, E. (2010). Climate change and mortality in Vienna—A human biometeorological analysis based on regional climate modeling. International Journal of Environmental Research and Public Health, 7, 2965–2977. https://doi.org/10.3390/ijerph7072965
- Petkova, E. P., Vink, J. K., Horton, R. M., Gasparrini, A., Bader, D. A., Francis, J. D., & Kinney, P. L. (2017). Towards more comprehensive projections of urban heat-related mortality: Estimates for New York city under multiple population, adaptation, and climate scenarios. *Envi*ronmental Health Perspectives, 125(1), 47–55. https://doi.org/10.1289/ehp166
- Porter, W. C., & Heald, C. L. (2019). The mechanisms and meteorological drivers of the summertime ozone-temperature relationship. Atmospheric Chemistry and Physics, 19(21), 13367–13381. https://doi.org/10.5194/acp-19-13367-2019
- Ren, C., Williams, G. M., Morawska, L., Mengersen, K., & Tong, S. (2008). Ozone modifies associations between temperature and cardiovascular mortality: Analysis of the NMMAPS data. Occupational and Environmental Medicine, 65(4), 255–260. https://doi.org/10.1136/ oem.2007.033878
- Rocque, R. J., Beaudoin, C., Ndjaboue, R., Cameron, L., Poirier-Bergeron, L., Poulin-Rheault, R. A., et al. (2021). Health effects of climate change: An overview of systematic reviews. *BMJ Open*, 11(6), e046333. https://doi.org/10.1136/bmjopen-2020-046333
- Rohat, G., Wilhelmi, O., Flacke, J., Monaghan, A., Gao, J., van Maarseveen, M., & Dao, H. (2021). Assessing urban heat-related adaptation strategies under multiple futures for a major US city. *Climatic Change*, *164*(3–4), 1–20. https://doi.org/10.1007/s10584-021-02990-9
- Roy, B., Khan, M. S. M., Islam, A. S., Mohammed, K., & Khan, M. J. U. (2021). Climate-induced flood inundation for the Arial Khan River of Bangladesh using open-source SWAT and HEC-RAS model for RCP8. 5-SSP5 scenario. SN Applied Sciences, 3(6), 648. https://doi.org/10.1007/s42452-021-04460-4

- Samet, J. M., Zeger, S. L., Dominici, F., Curriero, F., Coursac, I., Dockery, D. W., et al. (2000). The national morbidity, mortality, and air pollution study Part II: Morbidity and mortality from air pollution in the United States [Dataset]. Research Report Health Eff Inst, *94*, 5–79. Retrieved from https://www.healtheffects.org/publication/national-morbidity-mortality-and-air-pollution-study-part-ii-morbidity-and-mortality-air
- Vardoulakis, S., Dear, K., Hajat, S., Heaviside, C., Eggen, B., & McMichael, A. J. (2014). Comparative assessment of the effects of climate change on heat-and cold-related mortality in the United Kingdom and Australia. *Environmental Health Perspectives*, 122(12), 1285–1292. https://doi.org/10.1289/ehp.1307524
- Wang, Y., Shi, L., Zanobetti, A., & Schwartz, J. D. (2016). Estimating and projecting the effect of cold waves on mortality in 209 US cities. *Environment International*, 94, 141–149. https://doi.org/10.1016/j.envint.2016.05.008
- Weinberger, K. R., Haykin, L., Eliot, M. N., Schwartz, J. D., Gasparrini, A., & Wellenius, G. A. (2017). Projected temperature-related deaths in ten large US metropolitan areas under different climate change scenarios. *Environment International*, 107, 196–204. https://doi.org/10.1016/j. envint.2017.07.006
- Yang, J., Zhou, M., Ren, Z., Li, M., Wang, B., Liu, D. L., et al. (2021). Projecting heat-related excess mortality under climate change scenarios in China. Nature Communications, 12, 1–11. https://doi.org/10.1038/s41467-021-21305-1
- Yi, W., & Chan, A. P. (2015). Effects of temperature on mortality in Hong Kong: A time series analysis. *International Journal of Biometeorology*, 59(7), 927–936. https://doi.org/10.1007/s00484-014-0895-4
- Zhang, Y., Li, C., Feng, R., Zhu, Y., Wu, K., Tan, X., & Ma, L. (2016). The short-term effect of ambient temperature on mortality in Wuhan, China: A time-series study using a distributed lag non-linear model. *International Journal of Environmental Research and Public Health*, 13(7), 722. https://doi.org/10.3390/ijerph13070722

### **References From the Supporting Information**

- Gasparrini, A. (2014). Modeling exposure-lag-response associations with distributed lag non-linear models. *Statistics in Medicine*, 33(5), 881-899. https://doi.org/10.1002/sim.5963
- Gasparrini, A., Armstrong, B., & Kenward, M. G. (2012). Multivariate meta-analysis for non-linear and other multi-parameter associations. *Statistics in Medicine*, *31*(29), 3821–3839. https://doi.org/10.1002/sim.5471